Author’s response to reviews

Title: Assessing upper limb function: transcultural adaptation and validation of the Portuguese version of the Stroke Upper Limb Capacity Scale

Authors:

João Paulo Branco (joaobrancofmuc@hotmail.com)
Sandra Oliveira (sandra.oliveira1985@hotmail.com)
João Páscoa Pinheiro (reabmedica@hotmail.com)
Pedro Ferreira (pedrof@fe.uc.pt)

Version: 3 Date: 20 Jul 2017

Author’s response to reviews:

Authors Reply to Reviewers' and Editor’s comments

1. In your statement on Ethics and consent to participate please include the full names of the ethics committees (and the institutes to which they belong) that approved the study and the committees’ reference numbers if appropriate.

Please remove the heading “Aims” from the abstract and ensure you change this both in the manuscript and the submission information.

R – This information was now added to the manuscript. The heading was changed.

2. Please change the headings “Introduction” to “Background” and “Materials and Methods” to “Methods”

R – These headings were updated.

3. Please have the text checked by a native English speaking colleague. There are some issues with grammar, wording, spelling, and/or punctuation that need to be addressed. For example:

Page 3 “The majority of strokes was of ischaemic aetiology” should read “The majority of strokes WERE of ischaemic aetiology”
Page 5 “A total of 122 patients accepted to participate in the study, most of which had permanent functional disability” should read “A total of 122 patients AGREED to participate in the study, most of WHOM had permanent functional disability”

R – The paper was checked and corrections were made where appropriate.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

R – We submit only the clean versions as requested.